Author's response to reviews

Title: Array based characterization of a terminal deletion involving chromosome subband 15q26.2: An emerging syndrome associated with growth retardation, cardiac defects and developmental delay

Authors:

Josef Davidsson (josef.davidsson@med.lu.se)
Anna Collin (anna.collin@med.lu.se)
Gudrun Bjorkhem (gudrun.bjorkhem@skane.se)
Maria Soller (maria.soller@med.lu.se)

Version: 5 Date: 10 January 2008

Author's response to reviews: see over
Dear Editor

Please find our attached manuscript "Array based characterization of a terminal deletion involving chromosome subband 15q26.2: An emerging syndrome associated with growth retardation, cardiac defects and developmental delay" edited for publication in your journal.

We have now addressed the changes in the manuscript suggested by the Editorial Production team.

1. The statement regarding informed consent have been moved to the methods section
2. Authors Contributions have been changed to Authors’ contributions and it is indicated that all authors have read and approved the final version of the manuscript.
3. Reference 28 have been updated with publication details.
4. The website address for USCS Genome Browser now constitute reference 16, replacing the reference to the book chapter of Sambrook et al. that is now considered obsolete and do not need referral.

Yours Sincerely

Josef Davidsson

Corresponding Author:
Josef Davidsson, MSc, PhD student
Department of Clinical Genetics
Lund University Hospital
SE-221 85 Lund
Sweden
Tel +46 46 173398
Cell +46 709 544 807
Fax +46 46 131061
E-mail: Josef.Davidsson@med.lu.se